\documentclass[10pt,a4paper]{article}

\usepackage[latin1]{inputenc}
\usepackage{epsfig,amsfonts}
\usepackage{times}
\usepackage{geometry}
 \geometry{
 a4paper,
 total={210mm,297mm},
 left=30mm,
 right=30mm,
 top=30mm,
 bottom=30mm,
 }
%\usepackage{xcolor}
\setlength{\parindent}{5mm}
\pagestyle{empty}
%
\usepackage{graphicx}
\setlength\parindent{0pt}
\usepackage{booktabs}
\usepackage{hyperref}

\begin{document}

{\parindent0pt

{\Large Title of Presentation: Abstract Template for WECM'23}

\vspace{14pt}

First Author$^{1,2}$, Second Author$^{2}$, Third Author$^{3,*}$

\vspace{10pt}
$^1$ Affiliation 1 (University), Department, Postal Address, City, Country\\
$^2$ Affiliation 2 (University), Department, Postal Address, City, Country\\
$^3$ Affiliation 3 (University), Department, Postal Address, City, Country\\
$^*$ Presenting Author: email@address.edu\\


  
\vspace{10pt}
\textbf{Keywords:} keyword 1, keyword 2, keyword 3 (max 3).

\vspace{10pt}
}

\renewcommand{\section}[2]{}%

\noindent
\textbf{Abstract 1} %
Abstracts submitted to the 2\textsuperscript{nd} \emph{Workshop on Experimental and Computational Mechanics -- WECM'23} (Pisa, Italy, 20-22 September 2023) shall be formatted according to the present instructions.

The main text of the Abstract shall be placed here. Figures, Equations, and Tables can be included: some examples follow.

\begin{figure}[h]
\centering
\includegraphics[width=5cm]{example-image-a}
\caption{The figure description}
\label{fig_1} 
\end{figure}

The reference to Figure \ref{fig_1}.

The area of the circle is 
\begin{equation}
  A = \pi r^2
\label{eq_1}
\end{equation} 
where $r$ is the radius. Some results are shown in Table \ref{tab_1}. 

\begin{table}[h]
\centering
\begin{tabular}{cc}
  \toprule
  Radius (mm) & Area (mm$^2$) \\
  \midrule
  2.00 & 12.57\\
  4.00 & 50.27\\
  8.00 & 201.06\\
  \bottomrule
\end{tabular}
\caption{Some results from Eq. (\ref{eq_1})}
\label{tab_1} 
\end{table}

References should be recalled in the Abstract \cite{1}, where appropriate \cite{2,3}.\\

The final PDF should fit within one page in A4 paper size. Please, use the MS Word or LaTeX templates provided on the WECM'23 website: \url{http://wecm23.ing.unipi.it/}\\

\textbf{Acknowledgements:} Place here any acknowledgements (optional).

\vspace{14pt}
\textbf{References}
% \fontsize{8}{9}\selectfont
\begin{thebibliography}{99}
\setlength{\itemsep}{-2pt}
\bibitem{1} Red, A.B.; Orange, C.; Yellow, D.: \emph{Book Title}, Publisher, City (year). DOI: \dots%
\bibitem{2} Green, E.: Journal Paper Title, \emph{Journal Name}, 1 (2), 123:45--67 (year). DOI: \dots%
\bibitem{3} Blue, F.G.; Indigo, H.: Conference Paper Title, \emph{Conference Name} (City, Country, Date). DOI: \dots%
\end{thebibliography}

\end{document}
